food & nutrition (

REVIEW ARTICLE Vitamin D - a scoping review for Nordic nutrition recommendations 2023

Magritt Brustad^{1,3*} and Haakon E. Meyer^{2,4}

¹Department of Community Medicine, University of Tromsø – The Arctic University of Norway, Tromsø, Norway; ²Department of Community Medicine and Global Health, Institute of Health and Society, University of Oslo, Oslo, Norway; ³The Public Dental Health Service Competence Centre of Northern Norway (TkNN), Tromsø, Norway; ⁴Norwegian Institute of Public Health, Oslo, Norway

Popular scientific summary

- Vitamin D has important roles in calcium and phosphorous metabolism, as well as the development and maintenance of a healthy skeleton.
- Deficiency can cause rickets in infants and osteomalacia in adults/older people.
- Vitamin D can be obtained from both dietary sources and via sun exposure.
- Circulating 25-hydroxyvitamin D (25(OH)D) is the most reliable indicator of vitamin D status.
- Vitamin D may slightly reduce overall mortality and cancer mortality, but supplementation may not provide additional health benefits when vitamin D levels are already sufficient.

Abstract

Vitamin D is an essential nutrient. Its role in calcium and phosphorous metabolism, and in the development and maintenance of a healthy skeleton is well documented. In addition, there is some evidence for vitamin D decreasing total mortality and cancer mortality modestly, but not cancer incidence. Vitamin D is unique, as both diet and sun induced production in skin are sources to this vitamin. Individual vitamin D status is thus a sum of both sun exposure and dietary intakes. The discovery of vitamin D receptors and the activation of biological active vitamin D in numerous tissues and organs in the body has given support to hypothesis on vitamin D having extra-skeletal functions. The scientific literature on vitamin D and several health outcomes is high in numbers and has been increasing exponentially the last two decades. However, despite this large body of scientific publications and improvement in study quality, vitamin D supplementation has not shown to give additional health benefits when status is in sufficient range (i.e. circulating 25 hydroxyvitamin D >50 nmol/L). Well-designed studies on insufficient or deficient individuals are lacking.

The totality of evidence does not support that increased intake of vitamin D beyond current recommendation will have additional beneficial health effects.

Keywords: vitamin D; cholecalciferol; ergocalciferol; recommendations; NNR23

To access the supplementary material, please visit the article landing page

Received: 10 June 2022; Revised: 15 November 2022; Accepted: 18 September 2023; Published: 13 November 2023

itamin D_3 (cholecalciferol) is a steroid-like molecule synthesised from 7-dehydrocholesterol in the skin by ultraviolet B (UVB) light from the sun (wavelength between 290 and 315 nm) (1). Humans' requirement for vitamin D_3 can be met by exposing the skin to the sun radiation within this wavelength range. However, all the Nordic and Baltic countries $(54-71^{\circ}N)$ are situated at latitudes where the sun radiation is not sufficient part of the year for vitamin D₃ production in skin to occur (2). This time of the year is often referred to as the 'vitamin D winter'. The duration of the vitamin D winter is increasing by increasing latitude.

Food sources of vitamin D_3 are fish and seafood especially fatty fish like salmon, trout, mackerel, and herring. Egg yolk also contains vitamin D_3 . Some foods contain the metabolite 25-hydroxyvitamin D [25(OH)D], in addition to their cholecalciferol (vitamin D_3) content (like eggs), but there is no consensus on how to calculate the total contribution of vitamin D in such foods (3). Dairy products (including milk, butter and margarine) are fortified to a various degree in the Nordic countries. During recent years, the use of vitamin D fortification in different foods like oil and plant-based alternatives to milk products has increased, but it differs between countries. More details can be found in Itkonen et al. (4).

Vitamin D_2 (ergocalciferol) is a form of vitamin D used in some supplements and fortified foods. Some mushrooms like chanterelle are natural sources of D_2 . There is some evidence for D_3 being more potent than D_2 in raising vitamin D status in humans; however, inconsistency in results from studies assessing this has been reported (5).

Vitamin D is regarded as a pro-hormone. After entering the body, it is first converted (hydroxylated) to 25-hydroxyvitamin D [25(OH)D] in the liver. Thereafter it is further hydroxylated to the active form of vitamin D, 1,25-dihydroxyvitamin D (calcitriol), predominantly in the kidneys but also in other tissues. Circulating 25(OH) D is considered to be the most reliable biomarker for vitamin D-status in humans as it captures both dietary intake and cutaneous vitamin D-production. Consensus on cutoffs for defining biomarker-levels for 'sufficient', 'insufficient' and 'deficient', has been somehow hard to reach. However, based on available evidence there is a growing agreement that circulating 25(OH)D above 50 nmol/L corresponds to sufficient level, and less than 25–30 nmol/L indicates deficiency (5–8).

Vitamin D deficiency can occur if the diet is devoid of the vitamin and there is little or no exposure to UVB radiation. Infants can develop rickets and adults/elderly people can develop osteomalacia, and for this reason vitamin D is considered an essential micronutrient. The role of vitamin D for curing rickets in children, in mineralisation of bone, and calcium and phosphorous homeostasis is well described (6). Due to the role of vitamin D in securing a healthy skeleton, studies on vitamin D and the prevention of osteoporosis and fractures are many; however, most intervention studies are on vitamin D in combination with calcium.

Possible, so called 'extra-skeletal' effects of vitamin D on human health have gained increased research-attention during the last two decades as both the enzymes for activation of vitamin D and the vitamin D receptor are found in numerous tissues and organs in the body. For example, vitamin D is believed to play a role in various organ systems like muscle, brain, pancreatic β cells, and may potentially impact on the cardiovascular as well as the immune system (7). There are indications from e.g. genetic and molecular studies that vitamin D has various extra-skeletal effects (9). Many observational studies have shown associations between vitamin D and a long list of non-skeletal health outcomes and conditions.

The Nordic Nutrition Recommendations (NNR) from 2012 recommendation for vitamin D was based on a systematic literature review by Lamberg-Allardt et al. on quality assessed available evidence (7). This work concluded that there was an overall large heterogeneity in the literature, but conclusive evidence for protective effects on bone health, total mortality, and the risk of falling. It was emphasized that most intervention studies leading to these conclusions reported that intervention with vitamin D combined with calcium and not vitamin D alone, gave these benefits. Due to limited number of high quality randomized controlled trials available, the so-called strength of evidence (SOE) for some of the suggested health outcomes, was weak. Between the former NNR and NNR2023, there has been large increase in total number of scientific studies on vitamin D and health, also with some methodological improvements.

As a basis for setting DRVs, the aim of this scoping review is to describe updated evidence concerning the role of vitamin D on health-related outcomes since the previous version of NNR (NNR2012) (Box 1). Possible effects

Box 1. Context and process

- This paper is one of many scoping reviews commissioned as part of the Nordic Nutrition Recommendations 2023 (NNR2023) project (10).
- The papers are included in the extended NNR2023 report but, for transparency, these scoping reviews are also published in Food & Nutrition Research.
- The scoping reviews have been peer reviewed by independent experts in the research field according to the standard procedures of the journal.
- The scoping reviews have also been subjected to public consultations (see report to be published by the NNR2023 project).
- The NNR2023 committee has served as the editorial board.
- While these papers are a main fundament, the NNR2023 committee has the sole responsibility for setting dietary reference values in the NNR2023 project.

of vitamin D in the general population are considered, not for specific patient groups.

Methods

This review follows the protocol developed within the NNR2023 (10). The sources of evidence used in this scoping review follow the eligibility criteria described in the paper 'The Nordic Nutrition Recommendations 2022 – Principles and methodologies' published in Food & Nutrition Research (11).

Due to the very large amount of literature available on vitamin D and health, the literature search strategy was limited to 'umbrella reviews' or 'review of reviews'. This research strategy was decided in dialog with the NNR2023 Committee.

With assistance and advice from the Head Librarian at The University Library, UiT The Arctic University of Norway (Grete Overvåg), the search was developed. The full search string is found in Appendix A. The time period for the search was from January 2011 throughout 22 October 2021.

Proper quality assessment instruments are lacking for umbrella reviews, and an adapted version of AMSTAR 2 (Appendix B) was therefore developed and used in this review. Not all questions in AMSTAR 2 are relevant for evaluating the quality of umbrella reviews. The original AMSTAR 2 consists of 16 questions. Our adapted AMSTAR 2 contained the following original questions: 2.3.4.5.7.8.13.14, and 16 (confer Appendix B for details) as these were considered relevant for umbrella reviews. Question #3 was, however, slightly adapted as it was evaluated as fulfilled if the selection was 'reported and/or justified'. Thus, 'explanation for selection of study design for inclusion' as it is stated in the original # 3 question in the AMSTAR 2 tool, was not required. The #4 (literature search strategy used) was mandatory. Our adapted evaluation tool also included an additional final column where evaluation tool used in the included umbrella reviews was reported. The final grading of quality was based on a discretionary assessment, as the actual AMSTAR scores were not possible to calculate. The results from the included umbrella review were summarized in an evidence table (Supplementary Table 1).

Both the selection and quality assessment were done individually by the two authors and discordance discussed to reach agreements. In addition to the results from the systematic search and selection of umbrella reviews, qualified SRs (qSRs) identified by the NNR2023 Committee (5–7, 12, 13) constituted the source of evidence in this work. We have also consulted the UK Scientific Advisory Committee on Nutrition (SACN) publication from December 2020; i.e. published after the identification of qualified review (2019) by the NNR2023 Committee, entitled *Update of rapid review: Vitamin D and respiratory tract infections* (14). The remaining qSRs defined by the NNR2023 Committee were not included as final sources, as they were considered too old or newer updated versions were identified in the mentioned systematic search.

This review does not include an assessment of the dose-response relation between vitamin D intakes and circulating 25(OH)D concentrations. This is covered in Appendix 7 of the NNR2023 report (10).

Physiology

Vitamin D is the generic term for both vitamin D_3 (cholecalciferol) and vitamin D_2 (ergocalciferol). They are formed from their respective provitamins ergosterol and 7-dehydrocholesterol (7-DHC) upon UVB radiation exposure and subsequent thermal isomerisation. Vitamin D_2 differs from vitamin D_3 in the side chain where it has a double bond between C22 and C23 and an additional methyl group on C24.

Metabolism

The liver rapidly takes up vitamin D formed in the skin or absorbed from the gut where it is hydroxylated to 25(OH) D (Fig. 1). This metabolite is transported in plasma bound to the vitamin D binding protein (also known as the group-specific protein, Gc). The metabolite 25(OH) D is further converted into 1,25-dihydroxyvitamin D (calcitriol) in the kidneys. This is a calcium-regulating hormone that becomes active after binding to a nuclear vitamin D receptor. Together with parathyroid hormone and calcitonin, 1,25-dihydroxyvitamin D ensures that the concentration of calcium and phosphate in the plasma is maintained within narrow limits. Its main function is to stimulate the absorption of calcium from the intestine. In concert with parathyroid hormone, it also stimulates release of calcium from bone thereby increasing the concentration of circulating calcium. Deficiency of vitamin D may result in defective mineralization leading to the development of rickets in children and osteomalacia in adults (7).

The presence of both enzymes for activation of vitamin D and the vitamin D receptors in several tissues, as well as epidemiological and experimental data indicate that vitamin D might also have extra-skeletal roles.

Uptakes and bioavailability

There is evidence for both increased absorption of vitamin D from oral intakes and higher level of skin production in people with deficient vitamin D status compared to sufficient status (6). There is some evidence that the bioavailability differs between different food sources (6, 7).

Due to low content of vitamin D in breast milk, infants are recommended to receive vitamin D supplementation in all Nordic countries. In the former NNR,



Fig. 1. Vitamin D metabolism (6).

the vitamin D recommendation has been equal for children, adolescents, pregnant or lactating women, and adults. Vitamin D status remains stable during pregnancy (5). Due to reduced skin production capacity by age and reduced absorption for elderly, recommendation is higher for those >75 years.

There are some indications in the literature (15) that intake of an equal dose of vitamin D may give different 25(OH)D levels, due to different genetic profiles (polymorphisms). However, the practical implications of this are unclear.

Assessment of nutrient status

Plasma or serum concentration of 25(OH)D represents total vitamin D from diet and cutaneous synthesis. It is the most reliable biomarker for vitamin D status. However, wide-spread method-related discrepancies between different laboratories have caused concerns, especially when comparing

data from different studies and laboratories. As a response to this problem, several efforts have been made to standardize results across laboratories. This includes the extensive Vitamin D Standardization Program (VDSP), which also has been used to standardize 25(OH)D measurements from population-based studies in the Nordic countries (16).

The liquid-chromatography-tandem-mass-spectrometry (LC-MS/MS) is considered the most valid method for measurement of vitamin D metabolites. The 25(OH)D metabolite has a mean half-life in blood at around 2–3 weeks. Despite laboratory assay challenges, it is the established biomarker for both vitamin D2 and vitamin D3 (6). There is broad consensus that 25(OH)D levels below 25–30 nmol/L indicate vitamin D deficiency (5).

Dietary intake in Nordic and Baltic countries

In general, data on intakes in Europe has shown increased intakes by increasing latitude. However, there are large variations between the Nordic countries when it comes to level of fortification of vitamin D in foods, which in turn affects the actual intake. Finland is an example of an extensive vitamin D fortification program which has contributed to an improved vitamin D status in the population (4). Supplementation, especially in the form of cod liver oil, is also widely used, especially in Iceland and Norway.

Concerning infants, EU legislations require fortification of infant formula. In accordance with EU legislation, cereal-based products for infants/young children are fortified with vitamin D (4, 17).

The Nordic and Baltic counties are situated within the 'vitamin D winter window' where the UVB radiation from the sun is not sufficient for skin to produce vitamin D part of the year (2). This requires oral sources to the vitamin to secure sufficient vitamin D status in the population.

There are vulnerable groups in the population due to low intake or limited sun exposure. Skin pigmentation will also, to some degree, attenuate vitamin D production. At-risk groups include some immigrant groups from Asia and Africa, old people, infants, and teenagers. Strategies have been developed to increase intakes in line with recommendations in these groups (4).

Mean vitamin D intake (excluding supplements) has been reported to be markedly below the recommended intake in all Nordic and Baltic countries except for Finland (18). The higher intake in Finland is in line with their successful voluntary vitamin D fortification policy (18).

Based on selected studies, mean concentration of 25(OH)D seems to be in the sufficient range and approximately similar in the Nordic countries (60–65 nmol/L) among adults, except that older data indicate that the concentrations are lower in Iceland (4). Despite this, a notable proportion of the populations have concentration <50 nmol/L (vitamin D insufficiency), especially during winter. We refer to the Appendix 7 of the NNR2023 report (10) for dose-response discussions.

Health outcomes relevant for Nordic and Baltic countries

Results from systematic research of umbrella reviews

The selection procedure for umbrella reviews, reasons for exclusions, and final papers identified in the systematic search are all shown in the flow chart in Fig. 2.

The search in PubMed returned 25 hits. After independent assessment by two reviewers (MB and HEM) of title and abstract and discussion of discordances, nine umbrella reviews/review of reviews were selected for further evaluation. The identified papers were quality assessed by use of the adapted AMSTAR 2 checklist (Appendix B). Eight umbrella reviews were evaluated as being of sufficient quality, whereas one paper (19) was excluded because it was not a systematic review.

Health outcomes in umbrella reviews and qualified systematic reviews

Numerous health outcomes were included in the identified umbrella reviews. They are listed in Supplementary Table 1 (evidence table). Most of the included outcomes or topics in the identified umbrella reviews and qualified reviews were also included in the former NNR2012 process. However, there have been methodological improvements, increased number of studies and published original papers on these outcomes the last decade. New assessed outcomes identified in the umbrella reviews, or the qualified systematic reviews not mentioned in the former NNR2012 process, were: dementia/cognitive disorders, neuropsychological functioning, asthma, and pain. In addition, some of the outcomes mentioned in NNR2012 have now been more elaborated in the recent literature, including infection-related outcomes with specific infections/inflammation conditions as well as some immunomarkers and cancer mortality.

Conclusive evidence in NNR2012 - is it still valid?

In the systematic review which constituted the basis for the former NNR2012 recommendations on vitamin D by Lamberg-Allardt et al., it was concluded that vitamin D combined with calcium, but not vitamin D alone, reduced the risk of fracture (total fracture and hip fracture) (7).

The results from this present review of the umbrella reviews, confirm the lack of effect of vitamin D supplementation alone (without calcium) on the prevention of fractures in intervention studies. Challenges in the interpretation of the contribution of vitamin D in studies both administering vitamin D and calcium were already mentioned and discussed in the systematic review by Lamberg-Allardt et al. (7). The evidence for vitamin D not having



Fig. 2. Flow diagram of identification and inclusion of umbrella reviews. Source: Adapted from http://www.prisma-statement.org/.

an effect alone (without calcium) on preventing falls and fractures, has now been strengthened.

The recent literature has given increased evidence-weight for a small protective effect of vitamin D on total mortality. Three (20–22) of the identified umbrella reviews report vitamin D preventing total mortality with significant effect estimates around 6% or less. Autier et al. (21) reported that reduced mortality associated with vitamin D supplementation was not modified by concomitant calcium supplementation. The SACN vitamin D and health report from 2016 (5) reports reduced mortality risk for vitamin D and calcium combined.

Evidence-status for suggested relevant health outcomes

The hypotheses and list of other health outcomes and conditions possibly affected by vitamin D status are, to various degrees, supported by both biological- and observational evidence. However, updated systematic reviews including an increased number of randomized clinical trials (RCTs) have in general not shown clear beneficial preventing effects of vitamin D.

The umbrella reviews identified support a preventive effect on cancer mortality ranging from 12 to 16% reduction in risk estimates (21, 23), but not on cancer incidence.

Biological evidence for vitamin D's immunomodulatory and anti-inflammatory properties (24) supports the hypothesis on vitamin D having beneficial role on immune responses. In the identified umbrella reviews, we found some evidence supporting a preventive effect of vitamin D on acute respiratory tract infections (ARTI) (25). The Update of rapid review: Vitamin D and acute respiratory tract infections by SACN December 2020 (26) aimed at assessing evidence from RCTs on vitamin D and risk of ARTIs published after the 2016 SACN report on Vitamin D and Health (5). The SACN 2020 (26) concluded that vitamin D may reduce the risk of respiratory tract infection, but that the size of any potential benefit of vitamin D in reducing acute RTI risk may be small. Further, the SACN 2020 report (26) also concludes that evidence does not support vitamin D supplementation as preventive means for ARTIs, due to large degree of heterogeneity both methodologically with regards to study settings, doses, reporting and assessment outcomes, as well as the fact that RCTs published after 2017 did overall not report preventive effect of vitamin D on ARTIs (5).

Three (20, 27, 28) out of the eight identified umbrella reviews encompassed pregnancy outcome related topics. Strongest support was articulated by Matteussi et al. (20) based on their review on Cochrane systematic reviews where they conclude with 'some benefits from vitamin D supplementation' on preterm birth risk and low birth weight with a risk reduction at 64 and 60% respectively. However, this was not supported in the two other identified umbrella reviews (27, 28) mainly due to low quality evidence. Due to low methodological quality, conclusions could not be made on an effect of vitamin D on cognition and dementia (25, 27, 29).

Toxicity

Hypercalcemia, bone demineralization, calcification of soft tissue and renal damage are reported as outcomes of acute and chronic exposure to very high vitamin D intakes. Hypercalcemia has been defined as the most accurate endpoint (5). Adverse effects of excess vitamin D were not the topic for any of the identified umbrella reviews; however, some RCTs have reported increased risk for some outcomes under study (falls and fractures) in the intervention group after high dose vitamin D supplementation given as very large bolus doses (5).

The European Food Safety Authority (EFSA) 2016 (6) states that circulating 25(OH)D concentrations above 220 nmol/L may lead to hypercalcemia. The qualified review from EFSA, *Update of the tolerable upper intake level for vitamin D for infants* (12), concludes with upper limit (UL) of 25 μ g/day for infants 0–6 months, and a UL of 35 μ g/day for infants 6–12 months.

The UL in NNR2012 at an intake of 100 μ g/d, was based on the conclusion by EFSA in 2012 (30) and Institute of Medicine (IOM) (31) in 2011. The identified literature for this review, does not provide additional evidence for revising this former UL.

Requirement and recommended intakes

The potential disease prevention role of vitamin D has gained much attention in the past decade. This attention has resulted in a large increase in original papers as well as reviews. During this decade, the total scientific knowledge in the field has moved upwards in the evidence hierarchy, from observational evidence to increasing number of RCTs, meta-analysis, and systematic reviews.

Overall, the summarized evidence extracted from the large amount of literature shows that as methodological quality increases, the evidence has become weaker for a preventive effect of vitamin D on most outcomes.

There is observational support for associations between vitamin D status and health on various health outcomes. However, the interpretation for associations has challenges due to various biases and confounding. On the other hand, one could argue that RCTs have some methodological limitations in favor for observational studies. This is because RCTs on vitamin D cannot necessarily capture the sufficient long-time exposure of vitamin D relevant for some diseases, the control group is never a 'non-exposed group', as well as undefined dose-response curves causing inappropriate designs with regards to intervention-doses and baseline-levels. On the other hand, it is also well known that vitamin D status is associated with other risk factors for disease like physical activity or fish intakes (7). In addition, circulating 25(OH)D might be influenced by disease processes, and reverse causation may occur (5).

Bone health including falls

Vitamin D's role in preventing rickets and osteomalacia is well established. In the different revisions on dietary recommendations during the last decades, intakes to control these diseases have not been a pronounced issue when setting recommendations, as preventing these diseases is believed to require only modest doses of vitamin D (5).

Bone health in addition to rickets and osteomalacia was the main determinant for the recommendations in IOM (32) and in the systematic review for the NNR2012 by Lamberg-Allardt et al. (7). The main challenge in interpreting the available literature on these endpoints has been the administration of both calcium and vitamin D in the intervention groups. In Lamberg-Allardt et al. (7), it was concluded that no fracture preventing effect of vitamin D alone has been shown in RCTs. As more studies have been conducted intervening with vitamin D alone, this statement has been strengthened.

Lamberg-Allardt et al. (7) concluded that there was 'overall fair evidence that vitamin D with calcium is effective in preventing falls in the elderly especially in those with low baseline 25(OH)D concentrations, both community dwelling and in nursing care facilities'. Limited new information was found in the identified umbrella reviews. Mateussi et al. (20) only included systematic reviews already included in Lamberg-Allardt et al. Theodoraout et al. (27) concluded that based on RCTs, vitamin D alone had no preventive effect on falls.

Based on available literature assessed in this review, there is little or no evidence that raising circulating 25(OH)D concentration above 50 nmol/L has any additional bone health impact.

Non-skeletal health outcomes

Hypothesis on non-skeletal health benefits from increased circulating vitamin D concentrations has caused the exponential increase in research on vitamin D and numerous health outcomes in the last two decades. There is strong biological evidence for vitamin D having a role beyond calcium metabolism and the mineralization of bone. The mechanistic and molecular knowledge about this is increasing quickly. However, randomized controlled trials have to a large extent failed to confirm health benefits of vitamin D supplementation (except for total mortality, and for cancer mortality where the evidence has been strengthened). On the other hand, methodological critique of these RCTs has focused on the fact that most studies have been conducted in subjects with circulating 25(OH)D concentration above 50 nmol/L, i.e. the level identified as sufficient. In addition, high doses of vitamin D have been administered, either at daily, monthly, or even at yearly intervals.

Interestingly, there are indications of effects from supplementations among subgroups with deficient levels of circulating 25(OH)D concentrations. The ethical challenges, however, with conducting RCTs on selected populations at deficient levels are obvious. With regard to setting recommended intakes for vitamin D, this indication of an effect of supplementation in deficient groups supports the aim of avoiding vitamin D deficiency in the population (5). However, the literature in the current review does not support that circulating 25(OH)D concentrations beyond 50 nmol/L is required to achieve sufficient vitamin D health.

Knowledge gaps

Methodological improvements, a large number of studies and published original papers have increased the total SOE. In particular, the growing number of well-designed RCTs on vitamin D and several outcomes have increased the SOE compared to the evidence status a decade ago. However, identified weaknesses are challenges related to calcium being administered together with vitamin D interventions, few studies conducted on participants with deficient 25(OH)D concentrations, and still lack of well-designed RCTs on some suggested vitamin D related health outcomes. More knowledge on vitamin D status being a result of, more than a cause of diseases and ill health, could have methodological implications on future study designs. In addition, more knowledge on the direction of the relation between vitamin D and diseases will have implications on the interpretation of available data. The role of vitamin D in inflammation and inflammation-related diseases is interesting. However, based on today's knowledge the evidence is weak and cannot guide dietary recommendations.

Implications for recommendations

There is convincing evidence for recommendations to be set to prevent the population from being vitamin D deficient, defined as 25(OH)D < 30 nmol/L. There is an increasing body of evidence showing that there is no additional health benefit from increasing the 25(OH)D levels above the suggested sufficient level at around 50 nmol/L.

Based on the totality of present available scientific evidence on vitamin D and health, the overall picture is in line with what was described a decade ago when the NNR2012 was set. The updated evidence does not justify revision of the recommendations. The SOE has increased due to the large research activity within this field. Thus, there is stronger certainty now to conclude that increasing the recommendations will not have an effect in reducing disease risks in the population.

References

- Engelsen O. The relationship between ultraviolet radiation exposure and vitamin D status. Nutrients 2010; 2(5): 482–95. doi: 10.3390/nu2050482
- Engelsen O, Brustad M, Aksnes L, Lund E. Daily duration of vitamin D synthesis in human skin with relation to latitude, total ozone, altitude, ground cover, aerosols and cloud thickness. Photochem Photobiol 2005; 81(6): 1287–90. doi: 10.1562/2004-11-19-RN-375
- Fødevarestyrelsen. Fødevaredatabanken Version 4.2 Juni 2022. Ministreit for Fødevarer, Landbruk og Fiskeri: Fødevareinstituttet, Danmarks Teksniske Universitet; 2022. Available from: https:// frida.fooddata.dk/pdf/da-frida-4.2-dokumentation.pdf [cited 14 October 2023].
- Itkonen ST, Andersen R, Bjork AK, Brugard Konde A, Eneroth H, Erkkola M, et al. Vitamin D status and current policies to achieve adequate vitamin D intake in the Nordic countries. Scand J Public Health 2021; 49(6): 616–27. doi: 10.1177/1403494819896878
- 5. SACN. SACN vitamin D and health report. London: gov.uk; 2016.
- EFSA. Vitamin D: EFSA sets dietary reference values. World Food Regul Rev 2016; 26(6): 15.
- Lamberg-Allardt C, Brustad M, Meyer HE, Steingrimsdottir L. Vitamin D – a systematic literature review for the 5th edition of the Nordic nutrition recommendations. Food Nutr Res 2013; 57: 22671. doi: 10.3402/fnr.v57i0.22671
- Bouillon R. Comparative analysis of nutritional guidelines for vitamin D. Nat Rev Endocrinol 2017; 13(8): 466–79. doi: 10.1038/nrendo.2017.31
- Bouillon R, Marcocci C, Carmeliet G, Bikle D, White JH, Dawson-Hughes B, et al. Skeletal and extraskeletal actions of vitamin D: current evidence and outstanding questions. Endocr Rev 2019; 40(4): 1109–51. doi: 10.1210/ er.2018-00126
- Blomhoff R, Andersen R, Arnesen EK, Christensen JJ, Eneroth H, Erkkola M, et al. Nordic nutrition recommendations 2023. Copenhagen: Nordic Council of Ministers; 2023.
- Christensen JJ, Arnesen EK, Andersen R, Eneroth H, Erkkola M, Høyer A, et al. The Nordic nutrition recommendations 2022 – principles and methodologies. Food Nutr Res 2020; 64: 4402. doi: 10.29219/fnr.v64.4402
- Turck D, Bresson JL, Burlingame B, Dean T, Fairweather-Tait S, Heinonen M, et al. Update of the tolerable upper intake level for vitamin D for infants. EFSA J 2018;16(8):e05365–n/a. doi: 10.2903/j.efsa.2018.5365
- Newberry SJ, Chung M, Shekelle PG, Booth MS, Liu JL, Maher AR, et al. Vitamin D and calcium: a systematic review of health outcomes (update). Evid Rep Technol Assess (Full Rep) 2014; 217: 1–420. doi: 10.23970/AHRQEPCERTA217
- SACN. SACN rapid review: vitamin D and acute respiratory tract infections. 2020. Available from: https://www.gov.uk/government/publications/sacn-rapid-review-vitamin-d-and-acute-respiratory-tract-infections [cited 14 October 2023].
- 15. Nissen J, Vogel U, Ravn-Haren G, Andersen EW, Madsen KH, Nexo BA, et al. Common variants in CYP2R1 and GC genes are both determinants of serum 25-hydroxyvitamin D concentrations after UVB irradiation and after consumption of vitamin D(3)-fortified bread and milk during winter in Denmark. Am J Clin Nutr 2015; 101(1): 218–27. doi: 10.3945/ajcn.114.092148
- Cashman KD, Dowling KG, Škrabáková Z, Kiely M, Lamberg-Allardt C, Durazo-Arvizu RA, et al. Standardizing serum 25-hydroxyvitamin D data from four Nordic population

samples using the vitamin D standardization program protocols: shedding new light on vitamin D status in Nordic individuals. Scand J Clin Lab Invest 2015; 75(7): 549–61. doi: 10.3109/00365513.2015.1057898

- 17. EU. Commission Delegated Regulation (EU) 2019/828 of 14 March 2019 amending delegated regulation (EU) 2016/127 with regard to vitamin D requirements for infant formula and erucic acid requirements for infant formula and follow-on formula 2019. Available from: https:// eur-lex.europa.eu/legal-content/EN/TXT/PDF/?uri= CELEX:32019R0828&from=EL [cited 14 October 2023].
- Warensjö Lemming E, Pitsi T. The Nordic nutrition recommendations 2022 – food consumption and nutrient intake in the adult population of the Nordic and Baltic countries. Food Nutr Res 2022; 66: 1–11. doi: 10.29219/fnr.v66.8572
- Ganmaa D, Enkhmaa D, Nasantogtokh E, Sukhbaatar S, Tumur-Ochir KE, Manson JE. Vitamin D, respiratory infections, and chronic disease: review of meta-analyses and randomized clinical trials. J Intern Med 2022; 291(2): 141–64. doi: 10.1111/joim.13399
- Mateussi MV, Latorraca COC, Daou JP, Martimbianco ALC, Riera R, Pacheco RL, et al. What do cochrane systematic reviews say about interventions for vitamin D supplementation? Sao Paulo Med J 2017; 135(5): 497–507. doi: 10.1590/1516-3180.2017.0230150817
- Autier P, Mullie P, Macacu A, Dragomir M, Boniol M, Coppens K, et al. Effect of vitamin D supplementation on non-skeletal disorders: a systematic review of meta-analyses and randomised trials. Lancet Diabetes Endocrinol 2017; 5(12): 986–1004. doi: 10.1016/S2213-8587(17)30357-1
- 22. Rejnmark L, Bislev LS, Cashman KD, Eiriksdottir G, Gaksch M, Grubler M, et al. Non-skeletal health effects of vitamin D supplementation: a systematic review on findings from meta-analyses summarizing trial data. PLoS One 2017; 12(7): e0180512. doi: 10.1371/journal.pone.0180512
- Sluyter JD, Manson JE, Scragg R. Vitamin D and clinical cancer outcomes: a review of meta-analyses. JBMR Plus 2021; 5(1): e10420. doi: 10.1002/jbm4.10420
- Chun RF, Liu PT, Modlin RL, Adams JS, Hewison M. Impact of vitamin D on immune function: lessons learned from genome-wide analysis. Front Physiol 2014; 5: 151. doi: 10.3389/fphys.2014.00151

- Maretzke F, Bechthold A, Egert S, Ernst JB, Melo van Lent D, Pilz S, et al. Role of vitamin D in preventing and treating selected extraskeletal diseases – an umbrella review. Nutrients 2020; 12(4): 1–36. doi: 10.3390/nu12040969
- 26. SACN. Update of rapid review: vitamin D and acute respiratory tract infections. London: gov.uk; 2020.
- Theodoratou E, Tzoulaki I, Zgaga L, Ioannidis JP. Vitamin D and multiple health outcomes: umbrella review of systematic reviews and meta-analyses of observational studies and randomised trials. BMJ 2014; 348: g2035. doi: 10.1136/bmj. g2035
- Bialy L, Fenton T, Shulhan-Kilroy J, Johnson DW, McNeil DA, Hartling L. Vitamin D supplementation to improve pregnancy and perinatal outcomes: an overview of 42 systematic reviews. BMJ Open 2020; 10(1): e032626. doi: 10.1136/ bmjopen-2019-032626
- 29. Aghajafari F, Pond D, Catzikiris N, Cameron I. Quality assessment of systematic reviews of vitamin D, cognition and dementia. BJPsych Open 2018; 4(4): 238–49. doi: 10.1192/bjo.2018.32
- EFSA Panel on Dietetic Products, Nutrition and Allergies (NDA); Scientific Opinion on the Tolerable Upper Intake Level of vitamin D. EFSA Journal 2012; 10(7): 2813. doi: 10.2903/j. efsa.2012.2813
- 31. Institute of Medicine Committee to Review Dietary Reference Intakes for Vitamin D, Calcium. The national academies collection: reports funded by National Institutes of Health. In: Ross AC, Taylor CL, Yaktine AL, Del Valle HB, eds. Dietary reference intakes for calcium and vitamin D. Washington, DC: National Academies Press (US); 2011, pp. 1–524.
- Rosen CJ, Abrams SA, Aloia JF, Brannon PM, Clinton SK, Durazo-Arvizu RA, et al. IOM committee members respond to endocrine society vitamin D guideline. J Clin Endocrinol Metab 2012; 97(4): 1146–52. doi: 10.1210/jc.2011-2218

*Magritt Brustad

Department of Community Medicine University of Tromsø – The Arctic University of Norway Tromsø, Norway Email: magritt.brustad@uit.no Magritt Brustad and Haakon E. Meyer